Author's response to reviews

Title: An unusual presentation of multiple cavitated lung metastases from colon carcinoma.

Authors:

Patrizia Seminara (patrizia.seminara@uniroma1.it)  
Gaia Manna (gaia.manna@hotmail.it)  
Alessandra Emiliani (emiliani.alessandra@libero.it)  
Alessandro Iannace (ale_iann@hotmail.com)  
Tania Losanno (tania.los@libero.it)  

Version: 5  
Date: 27 May 2011

Author's response to reviews:

Cover letter 27.05.2011

In reply to your major revision, we made the following changes (as required):

1. Affiliations: all authors are from the same Department

2. Abstract and Structure: We reformatted them, using the indicated headings.

3. Figures: Each page has only one legend:

On page 10 there is the legend of Figure 1 (a) and (b)

On page 11 the legend of Figure 2